RESEARCH





Response and progression-free survival according to planned treatment duration in patients with relapsed multiple myeloma treated with carfilzomib, lenalidomide, and dexamethasone (KRd) versus lenalidomide and dexamethasone (Rd) in the phase III ASPIRE study

Meletios Dimopoulos^{1*}, Michael Wang², Vladimir Maisnar³, Jiri Minarik⁴, William Bensinger⁵, Maria-Victoria Mateos⁶, Mihaela Obreja⁷, Julie Blaedel⁷ and Philippe Moreau⁸

Abstract

Background: In ASPIRE, carfilzomib, lenalidomide, and dexamethasone (KRd) significantly improved progression-free survival (PFS) and response rates versus lenalidomide and dexamethasone (Rd) in patients with relapsed multiple myeloma. Per protocol, patients received KRd for a maximum of 18 cycles followed by Rd to progression, so the benefit/ risk profile of KRd to progression was not established.

Methods: This post hoc analysis evaluated the efficacy and safety of KRd versus Rd at 18 months from randomization. Cumulative rates of complete response (CR) or better over time and PFS hazard ratio (HR) at 18 months were evaluated for KRd versus Rd. PFS HRs were also assessed according to cytogenetic risk, prior lines of therapy, and prior bortezomib treatment. Cox regression analysis was used to evaluate PFS HRs.

Results: The hazard ratio (HR) for PFS at 18 months was 0.58 versus 0.69 for the overall ASPIRE study. Patients with high-risk cytogenetics, ≥ 1 prior lines of therapy, and prior bortezomib exposure benefited from KRd up to 18 months versus Rd. The HRs for PFS at 18 months in the pre-defined subgroups were lower than those in the overall study. The difference in the proportion of KRd and Rd patients achieving at least a complete response (CR) increased dramatically over the first 18 months and then remained relatively constant. The safety profile at 18 months was consistent with previous findings.

Conclusions: The improved PFS HR at 18 months and the continued increase in CR rates for KRd through 18 cycles suggest that there may be a benefit of continued carfilzomib treatment.

Trial registration: Clinical trials.gov NCT01080391. Registered 2 March 2010.

Keywords: Clinical research, Clinical trials, Multiple myeloma, Myeloma therapy

* Correspondence: mdimop@med.uoa.gr

Full list of author information is available at the end of the article



© The Author(s). 2018 **Open Access** This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated.

¹School of Medicine, National and Kapodistrian University of Athens, Athens, Greece

Background

Multiple myeloma (MM) is a hematologic malignancy characterized by clonal proliferation of plasma cells in the bone marrow [1]. According to Globocan cancer incidence and mortality statistics, the estimated incidence of MM worldwide in 2012 was 114,251, and approximately 80,019 deaths attributed to the disease were reported [2]. MM remains an incurable disease characterized by a recurring pattern of relapse and remission. Upon relapse, patients face poor outcomes, which worsen with each relapse as a response to treatment, duration of response, and health-related quality of life decline [3-7]. In addition, the cost of managing the disease increases [8-10]. Recent advances in the treatment of MM have led to improvements in depth of response, progression-free survival (PFS), and overall survival (OS) in patients with MM [11–13]. Despite improvements in clinical outcome, almost all patients with MM eventually relapse. With many treatment options available for MM patients, there is a need to understand the optimal use of these therapies to achieve an appropriate balance between efficacy and safety. Based on positive clinical study results showing improvements in PFS and OS, there has been a shift to administer continuous therapy for both transplant-eligible and transplant-ineligible MM patients in place of a fixed-dose treatment [14-20]. The rationale for continuous therapy is to control minimal residual disease in order to delay disease recurrence [21].

Carfilzomib is a potent, irreversible, and selective proteasome inhibitor that has shown robust activity in myeloma, both as a single agent and in combination with other anti-myeloma agents [22-26]. Of note, carfilzomib has shown favorable clinical outcomes in patients with relapsed and/or refractory MM with a lower incidence of peripheral neuropathy when compared head-to-head with bortezomib [22]. The dosing and administration schedule of carfilzomib have evolved as the drug has progressed through clinical studies. In the phase Ib/II study (PX-171-006) of carfilzomib in combination with lenalidomide and dexamethasone (KRd), carfilzomib was administered as a 10-min intravenous (IV) infusion initiated at 15 mg/m² and escalated to a maximal dose of 27 mg/m^2 in patients with relapsed MM [27]. The overall response rate was 77% in patients who received the maximum planned dose of carfilzomib (27 mg/m^2) , and adverse events were consistent with the known safety profiles of the three agents [27, 28]. Based on these findings, the carfilzomib dose of 27 mg/m² was selected for further evaluation in the ASPIRE study [25].

The ASPIRE study evaluated KRd in comparison with lenalidomide and dexamethasone (Rd) alone in patients with relapsed MM who had received one to three prior lines of therapy. The addition of carfilzomib to Rd resulted in significantly higher overall response rate versus Rd alone (87.1 vs 66.7%; p < 0.001) and longer PFS (26.3 vs Rd, 17.6 months; hazard ratio [HR] 0.69; 95% confidence interval [CI] 0.57–0.83) [25]. In addition, results from the final OS analysis of ASPIRE showed an almost 8-month improvement in OS in patients treated with KRd compared with Rd (48.3 vs 40.4 months; HR 0.79; 95% CI 0.67-0.95; p = 0.0045) [29]. Carfilzomib treatment was discontinued after 18 cycles because data on the long-term safety of carfilzomib were not yet available when the AS-PIRE study was initiated. Because carfilzomib was discontinued after 18 cycles, the optimal duration of KRd treatment was not determined. Herein, we performed a post hoc analysis of the ASPIRE study to compare the safety and efficacy of KRd versus Rd at 18 months from randomization (i.e., at the latest time that allowed a direct comparison of KRd vs Rd).

Methods

The ASPIRE study has been previously described in detail by Stewart et al. [25]. The primary endpoint of the study was PFS. Secondary endpoints included OS, overall response rate (partial response or better), and safety. Prior bortezomib therapy was allowed, provided that patients did not have disease progression during treatment. Prior Rd treatment was also permitted if patients had not progressed during the first 3 months of therapy or discontinued treatment due to intolerance. All patients were required to have adequate renal, hematological, and hepatic function at screening. Patients who had grade 3 or 4 peripheral neuropathy (or grade 2 with pain) within 14 days before randomization or New York Heart Association class III or IV heart failure were excluded. All patients provided informed consent, and the study was performed according to the Declaration of Helsinki. The study protocol was approved by the institutional review boards of all participating institutions.

Patients were randomized 1:1 to receive KRd or Rd in 28-day cycles until the withdrawal of consent, disease progression, or the occurrence of toxicity. The stratification factors used for randomization were prior bortezomib therapy, prior lenalidomide therapy, and β^2 microglobulin levels. Patients were administered carfilzomib on days 1, 2, 8, 9, 15, and 16 of cycles 1-12 and on days 1, 2, 15, and 16 of cycles 13-18. Carfilzomib was given as a 10-min infusion at a starting dose of 20 mg/m² on days 1 and 2 of cycle 1, and 27 mg/m² thereafter. In the KRd and Rd arms, lenalidomide (25 mg) was given on days 1-21, and dexamethasone (40 mg) was given on days 1, 8, 15, and 22. Carfilzomib was discontinued after cycle 18 after which all patients on the carfilzomib arm continued to receive Rd treatment until disease progression.

Cumulative rates of complete response or better (\geq CR) over time were evaluated from the start of the treatment for KRd versus Rd. The PFS HR at 18 months for the intent-to-treat population was evaluated using the piecewise Cox regression analysis [30]. An analysis of PFS within the carfilzomib arm among patients who achieved \geq CR compared with patients who achieved < CR was performed using the Simon-Makuch landmark transient state approach [31]. Patients were classified as \geq CR or < CR according to their best overall response (BOR) at the landmark time. The choice of landmark time was based on the median time to reach a BOR of \geq CR or better, which was 6.7 months for the KRd arm and 8.3 months for the Rd arm. PFS events noted by landmark time were excluded from the analysis. This landmark method was used to minimize the bias in favor of responders represented by the time required to reach the response.

PFS HRs were also evaluated in subgroups of patients in ASPIRE according to cytogenetic risk, prior lines of therapy and prior bortezomib treatment. All patients who received at least one dose of study treatment were evaluated for safety analyses. The International Myeloma Working Group Uniform Response Criteria were used to assess response and disease progression [32]. According to these criteria, \geq CR was defined as CR (i.e., negative immunofixation on the serum and urine, disappearance of any soft tissue plasmacytomas, and < 5% plasma cells in bone marrow) plus stringent CR (i.e., normal free light chain ratio and absence of clonal cells in bone marrow by immunohistochemistry or immunofluorescence) [32].

Results

The cutoff date for the primary PFS analysis in ASPIRE was June 16, 2014, after a median follow-up of over 30 months. A total of 792 patients were enrolled in the

intent-to-treat population, and 396 each were randomized to the KRd and Rd arms. As reported in the primary ASPIRE publication [25], patient demographic and baseline disease characteristics for the intent-to-treat population were generally balanced between treatment arms, including cytogenetic risk, baseline neuropathy, number of prior regimens, and prior therapies.

The overall PFS HR in ASPIRE for KRd versus Rd was 0.69 (95% CI 0.57-0.83). At 18 months from randomization, the PFS HR was 0.58 (95% CI 0.46-0.72) in favor of the KRd group (Table 1). PFS HRs at 18 months also favored KRd versus Rd in patients with high-risk cytogenetics (HR 0.56; 95% CI 0.31-0.99), standard cytogenetic risk (HR 0.54; 95% CI 0.37-0.80), one prior line of treatment (HR 0.58; 95% CI 0.41-0.82), two or more prior lines of treatment (HR 0.60; 95% CI 0.45-0.81), and prior bortezomib exposure (HR 0.59; 95% CI 0.45-0.78) (Table 1). HRs were numerically lower at 18 months from randomization compared with HRs for the same subgroups from the primary ASPIRE study [25]. In the overall ASPIRE study, a total of 126 (32%) patients in the KRd arm and 37 (9%) in the Rd arm achieved \geq CR, with a median time from treatment start to \geq CR of 6.7 months for KRd and 8.3 months for Rd. The cumulative \geq CR rates are presented in Fig. 1. Cumulative \geq CR rates increased for both KRd-treated and Rd-treated patients over the course of 30 months, but \geq CR rates were higher for KRd versus Rd across all time points. At 18 months from randomization, 28.6% of patients who received KRd versus 7.7% of patients who received Rd achieved \geq CR (Fig. 1). There was a marked separation between the \geq CR curves of KRd-treated and Rd-treated patients in the first 9 to 12 months of treatment with \geq CR rates rising rapidly in the KRd arm. The \geq CR curves of

Table 1 PFS HRs for the ASPIRE ITT population and select subgroups

	At 18 months from randomization	Overall ASPIRE study [25, 36, 46]		
	HR (KRd/Rd) (95% CI)	HR (KRd/Rd) (95% CI)		
Entire ASPIRE population	0.58 (0.46–0.72)	0.69 (0.57–0.83)		
Cytogenetic risk				
High ^a	0.56 (0.31–0.99)	0.70 (0.43–1.16)		
Standard	0.54 (0.37–0.80)	0.66 (0.48–0.90)		
Prior lines of treatment				
1	0.58 (0.41–0.82)	0.71 (0.53–0.96)		
2	0.65 (0.44–0.97)	0.75 (0.54–1.04)		
3	0.53 (0.34–0.84)	0.68 (0.47–1.00)		
Prior bortezomib treatment	0.59 (0.45–0.78)	0.70 (0.56–0.88)		
No prior bortezomib treatment	0.58 (0.39–0.86)	0.73 (0.52–1.02)		

CI confidence interval, HR hazard ratio, ITT intent-to-treat, KRd carfilzomib, lenalidomide, and dexamethasone, PFS progression-free survival, Rd lenalidomide and dexamethasone

^aHigh cytogenetic risk was defined by t(4;14), t(14;16), or del(17p) in \geq 60% of plasma cells. At baseline, 12.1% of KRd-treated patients and 13.1% of Rd-treated patients had high-risk cytogenetics



KRd- and Rd-treated patients continued to separate from 12 to 18 months and then remained approximately parallel to each other.

PFS rates were compared for KRd-treated patients who achieved \geq CR with KRd-treated patients who achieved < CR in the overall ASPIRE population after 6 months from randomization (Fig. 2). KRd-treated patients who achieved \geq CR were associated with higher PFS rates compared with KRd-treated patients who achieved < CR. Specifically, PFS rates for ≥ CR patients were 96, 87, 79, 67, and 55% at approximately 6, 12, 18, 24, and 30 months from the landmark (6 months from randomization), respectively, and PFS rates for < CR patients were 85, 68, 55, 43, and 36% at 6, 12, 18, 24, and 30 months from the landmark, respectively (Fig. 2). In the high-risk cytogenetics, prior bortezomib exposure, and one or more prior lines of treatment subgroups, KRd patients who achieved \geq CR were also associated with higher PFS rates at all observable time points from the landmark than KRd patients who achieved < CR (Fig. 3).

At 18 months from randomization, median treatment duration was 78 weeks (95% CI 1-80) in the KRd arm and 57 weeks (95% CI 1-79) in the Rd arm, and the



median number of cycles for KRd versus Rd was 19 (95% CI 1-20) versus 14 (95% CI 1-21). Disease progression was the most common reason for treatment discontinuation in the first 18 months of treatment, and treatment discontinuations due to disease progression were more frequent with Rd (36.2%) compared with KRd (20.9%). The rates of treatment-emergent adverse events, discontinuation, and deaths at 18 months from randomization are presented in Table 2. Rates of events (KRd vs Rd) were 96.7 versus 95.6% for any-grade adverse events and 80.4 versus 75.3% for grade ≥ 3 adverse events (Table 2). Serious adverse events occurred in 53.8% of patients treated with KRd and 46.3% of patients treated Rd. A total of 83 (21.2%) patients receiving KRd and 81 (20.8%) patients receiving Rd discontinued any study drug due to adverse events (Table 2). Deaths due to adverse events occurred in 25 (6.4%) KRd-treated patients and 29 (7.5%) Rd-treated patients (Table 2). Rates of any-grade thrombocytopenia (27.3 vs 21.6%) and grade \geq 3 thrombocytopenia (16.1 vs 12.1%) were higher for KRd versus Rd arm. The frequency of any-grade acute renal failure was similar between the KRd and Rd arms (KRd 6.6%; Rd 6.2%), but grade ≥ 3 acute renal failure occurred slightly more frequently in the Rd arm than the KRd arm (KRd 2.3%; Rd 3.1%). Any-grade and grade \geq 3 anemia and neutropenia



occurred slightly more frequently in the KRd arm than the Rd arm. The incidence of any-grade peripheral neuropathy was 15.1% for KRd and 14.1% for Rd. Grade \geq 3 peripheral neuropathy occurred at the same rate in the KRd and Rd arms (KRd 2.3%; Rd 2.3%). The incidence of cardiovascular events including any-grade hypertension (13.0 vs 5.9%), cardiac failure (5.9 vs 3.3%), and dyspnea (20.9 vs 17.2%) was higher for KRd versus Rd. Grade \geq 3 hypertension (4.1 vs 1.3%), cardiac failure (3.6 vs 1.3%), and dyspnea (2.6 vs 1.5%) occurred more frequently in the KRd arm than the Rd arm.

Discussion

Previously reported results from the phase III ASPIRE study showed that the addition of carfilzomib to Rd resulted in high \geq CR rates and a clinically meaningful and statistically significant improvement in PFS and OS in patients with relapsed MM who had received at least one to three prior lines of therapy [25, 29]. Importantly,

per protocol, carfilzomib application on days 8 and 9 was omitted after cycle 12, and carfilzomib treatment was completely discontinued after 18 cycles (Rd continued until progression). Notably, the PFS benefit observed in ASPIRE was not a result of poor outcomes in the Rd group as the reported median PFS in the Rd group was similar to that observed in other trials comparing triplet combinations to Rd [33–35]. For the primary PFS analysis, the median follow-up was 32.3 months in the KRd group and 31.5 months in the Rd group. An analysis of ASPIRE by previous treatment showed benefits in PFS with KRd versus Rd in patients with one prior line of therapy, \geq 2 previous lines of therapy, and previous bortezomib exposure over an approximately 3-year time period, including at 18 months [36].

In this post hoc analysis, we evaluated the efficacy and safety of KRd versus Rd at 18 months from randomization to enable a more robust evaluation of the addition of carfilzomib to Rd. We found that for the first 18 months

Table 2	Treatment-emergent	adverse events,	discontinuation	, and deaths at	18 months from	randomization	(safety pop	ulation)
---------	--------------------	-----------------	-----------------	-----------------	----------------	---------------	-------------	----------

	KRd (<i>n</i> = 392)	Rd (<i>n</i> = 389)
Patients with any-grade AE, n (%)	379 (96.7)	372 (95.6)
Grade \geq 3 AE, <i>n</i> (%)	315 (80.4)	293 (75.3)
Serious AEs, n (%)	211 (53.8)	180 (46.3)
AE leading to treatment discontinuation, n (%)	83 (21.2)	81 (20.8)
AE leading to death, n (%)	25 (6.4)	29 (7.5)
Any-grade AEs of interest, <i>n</i> (%)		
Anemia	157 (40.1)	147 (37.8)
Thrombocytopenia	107 (27.3)	84 (21.6)
Neutropenia	137 (34.9)	126 (32.4)
Hypertension	51 (13.0)	23 (5.9)
Dyspnea (HLT)	82 (20.9)	67 (17.2)
Peripheral neuropathy (SMQN)	59 (15.1)	55 (14.1)
Cardiac failure (SMQN)	23 (5.9)	13 (3.3)
Acute renal failure (SMQN)	26 (6.6)	24 (6.2)
Grade \geq 3 AEs of interest, <i>n</i> (%)		
Anemia	69 (17.6)	65 (16.7)
Thrombocytopenia	63 (16.1)	47 (12.1)
Neutropenia	111 (28.3)	99 (25.4)
Hypertension	16 (4.1)	5 (1.3)
Dyspnea	10 (2.6)	6 (1.5)
Peripheral neuropathy (SMQN)	9 (2.3)	9 (2.3)
Cardiac failure (SMQN)	14 (3.6)	5 (1.3)
Acute renal failure (SMQN)	9 (2.3)	12 (3.1)

AE adverse event, HLT high-level term, KRd carfilzomib, lenalidomide, and dexamethasone, Rd lenalidomide and dexamethasone, SMQN standardized Medical Dictionary for Regulatory Activities query, narrow scope

since randomization, deeper responses were achieved with KRd compared with Rd, and the PFS HR for KRd versus Rd was lower than that for the overall study, indicating a better treatment effect than that reported for the entire duration of the ASPIRE study. Specifically, treatment with KRd for 18 months reduced the risk of progression or death by 42%. The rate of increase in the proportion of patients achieving \geq CR was higher in the KRd arm compared with the Rd arm, particularly in the first 15 months from the start of treatment. It is known that a positive correlation exists between the depth of response and improved outcomes in terms of PFS and OS in patients with MM [37-39]. We found that the proportion of patients who reached \geq CR was approximately three to four times higher in the KRd than Rd arm (at 18 months from randomization, KRd 28.6%; Rd 7.7%). We compared PFS in KRd-treated patients who reached \geq CR with KRd-treated patients who achieved < CR. PFS rates in the KRd arm were higher for patients who reached \geq CR versus patients who achieved < CR at all observable time points in the overall ASPIRE study. These results show that PFS was improved in KRd-treated patients who achieved a deeper response.

We observed an increase in \geq CR rates in both KRd and Rd treatment arms, with further increase in \geq CR rates in the KRd arm beyond month 18 (Fig. 1). Patients with high disease burden may have achieved CR later compared with patients with smaller tumor burdens. In addition, there may have been subjective factors that could explain the delay in achieving deeper responses, including to the time the bone marrow tests were performed to confirm CR. For example, investigators may have delayed their decision to perform bone marrow tests based on their overall evaluation of the patient, or these tests were analyzed by a local lab and had to be redone by the central lab to be considered for inclusion in the clinical data.

In the relapsed setting, it is a common practice to continue treatment until progression, provided that the treatment is well-tolerated. In ASPIRE, treatment with lenalidomide and dexamethasone was continued until disease progression, whereas carfilzomib treatment was omitted after 12 cycles and discontinued after 18 cycles per protocol. In another randomized phase III studies such as TOURMALINE-MM1, ELOQUENT, and POL-LUX, with patient populations similar to the population in ASPIRE, treatment was continued until disease progression [33–35]. Based on our findings in this analysis, it can be speculated that if carfilzomib had been administered beyond 18 cycles, the PFS HR might have been improved further.

With respect to safety, the rates of treatment discontinuations and death due to adverse events were similar between the KRd and Rd arms at 18 months from randomization. Hematological adverse events including any-grade and grade \geq 3 anemia, thrombocytopenia, and neutropenia occurred more frequently in the KRd versus Rd arm at 18 months from randomization. The rates of any-grade and grade \geq 3 hypertension, dyspnea, and cardiac failure were also higher in patients receiving KRd than patients receiving Rd. The incidence of any-grade and grade \geq 3 peripheral neuropathy was similar between the KRd and Rd arms. Overall, the safety findings reported in this post hoc analysis were similar to those reported in the safety analysis of the primary ASPIRE study.

Treating MM can be a challenge, in particular for patients with high-risk cytogenetics. High-risk cytogenetics has been shown to have a negative impact on survival and prognosis [40-42]. In ASPIRE, high risk was defined by t(4;14), t(14;16), or del(17p) in $\ge 60\%$ of plasma cells. Of note, the advantage of KRd was retained across subgroups, including patients with high-risk cytogenetics at study entry. In addition, patients with relapsed and/or refractory disease present a challenge, because the benefit they receive from the therapy usually decreases with subsequent lines of therapy [43]. We found that in patients who had high-risk cytogenetics or one or more prior lines of therapy, PFS HRs favored KRd versus Rd in the first 18 months and were lower than the HR for the overall ASPIRE study [25]. As expected, PFS rates were higher for \geq CR responders than < CR responders in the KRd arm. Similar efficacy results were observed for the subgroup of patients who had prior bortezomib exposure, which is notable as many patients with relapsed MM will have been exposed to bortezomib [44, 45].

We observed that the PFS HR (0.58) for KRd versus Rd for the first 18 months was lower than the PFS HR for the overall ASPIRE study (0.69). The 18-month PFS HRs for patients with high or standard cytogenetic risk, one or two or more prior lines of treatment and prior, and no prior bortezomib exposure were also lower than those reported in the preplanned subgroup analyses of ASPIRE according to cytogenetic risk [46], prior lines of treatment [36], and prior therapy [36]. The lower HRs observed at 18 months are possibly due to patients receiving carfilzomib for a maximum of 18 months. Had patients received continuous carfilzomib treatment

in the ASPIRE study, we could have potentially observed even greater clinical benefit in the KRd arm, with longer PFS and deeper responses. Continuous treatment with KRd might have been even more efficacious in patients with high-risk cytogenetics. In fact, studies show that extended duration of therapy is associated with better clinical outcomes in patients with MM. Specifically, Palumbo et al. observed that newly diagnosed MM patients who received novel agentbased continuous therapy had longer PFS and improved OS than patients who had a fixed duration of therapy [19]. In addition, using data from the bortezomibmelphalan arm of the phase III VISTA study, Mateos et al. found that higher cumulative bortezomib dose, which reflects prolonged treatment duration, is associated with improved OS in newly diagnosed MM patients [47]. With respect to carfilzomib, newly diagnosed MM patients who received continuous treatment with KRd achieved high responses [48–50]. Of note, upon approval of KRd, European Medicines Agency concluded that treatment with KRd could continue beyond 18 cycles based on individual risk-benefit assessment, as the data on the tolerability and toxicity of carfilzomib beyond 18 cycles are limited [51]. Although we are not recommending routine administration of carfilzomib in the KRd regimen beyond 18 cycles, this analysis, despite the limitations of the post hoc design, supports further evaluation of carfilzomib treatment beyond 18 cycles. Additional larger studies are required to evaluate whether prolonged carfilzomib treatment duration leads to further improvements in clinical outcomes without additional adverse events.

Conclusions

This post hoc analysis of ASPIRE shows that KRd provided greater responses and improvements in PFS at 18 months compared with Rd alone. Cumulative \geq CR rates increased over time in the KRd arm and were higher than the cumulative \geq CR rates in the Rd arm, with rates rapidly rising in the first 15 months from the start of treatment. PFS HRs for the first 18 months from randomization were lower for KRd versus Rd than in the overall ASPIRE study and regardless of prior cytogenetic status, number of prior lines of treatment, and prior bortezomib exposure. Compared with the overall results of the ASPIRE study, the markedly lower HR for PFS at 18 months in this analysis suggests a potential additional clinical benefit of continued treatment with carfilzomib beyond 18 months. However, further evaluation is warranted.

Abbreviations

BOR: Best overall response; CR: Complete response; HR: Hazard ratio; IV: Intravenous; KRd: Carfilzomib, lenalidomide, and dexamethasone; MM: Multiple myeloma; OS: Overall survival; PFS: Progression-free survival; Rd: Lenalidomide and dexamethasone

Acknowledgements

All listed authors meet the criteria for authorship set forth by the International Committee for Medical Journal Editors. Medical writing support was provided by Jacqueline Sayyah, PhD, of Amgen Inc., and editorial assistance was provided by CACTUS Communications, Inc. and BlueMomentum, an Ashfield Company.

Funding

The ASPIRE study was supported by Onyx Pharmaceuticals Inc., an Amgen Inc. subsidiary.

Availability of data and materials

The primary ASPIRE analysis has been published (Stewart et al. New Engl J Med. 2015;372, 142–52), and data has been posted to Clinicaltrials.gov (https://clinicaltrials.gov/ct2/show/study/NCT01080391). The datasets used and/or analyzed during the current study are available from the corresponding author on request.

Authors' contributions

MD and PM designed the research study, acquired and analyzed the data, drafted the manuscript, and approved the final version. MW, VM, JM, WB, M-VM, and MO acquired and analyzed the data, drafted the manuscript, and approved the final version. JB analyzed the data, drafted the manuscript, and approved the final version.

Authors' information

Not applicable.

Ethics approval and consent to participate

All patients provided informed consent. The study was performed according to the Declaration of Helsinki. The study protocol, amendments, protocol clarification letters, informed consent forms, subject dosing diaries, advertisements, and health-related quality of life questionnaires were approved by the institutional review boards of all participating institutions.

Consent for publication

Not applicable.

Competing interests

MD—consultancy for Amgen, Celgene, Janssen, Novartis, and Takeda. MW—research funding from Pharmacyclics, Juno Therapeutics, Acerta Pharma, Celgene, BeiGene, Kite Pharma, Onyx, Asana BioSciences, and Janssen; consultancy for Acerta Pharma and Janssen. VM—grants from the Binding Site; personal fees from Amgen, BMS, Celgene, and Janssen-Cilag. JM—no conflicts of interest to disclose. WB—grants and research funding from Celgene and Amgen Inc. M-VM—personal fees from Amgen, Celgene, Janssen, and Takeda. MO and JB—employment and stock ownership in Amgen, Inc. PM—consultancy for Amgen, Bristol-Myers Squibb, Celgene, Janssen, Novartis, and Takeda.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Author details

¹School of Medicine, National and Kapodistrian University of Athens, Athens, Greece. ²The University of Texas MD Anderson Cancer Center, Houston, TX, USA. ³Faculty Hospital and Medical Faculty of Charles University in Hradec Kralove, Hradec Kralove, Czech Republic. ⁴University Hospital Olomouc, Medical Faculty of Palacky University Olomouc, Olomouc, Czech Republic. ⁵Fred Hutchinson Cancer Research Center, Seattle, WA, USA. ⁶Hospital Clinico Universitario de Salamanca-IBSAL, Salamanca, Spain. ⁷Amgen, Inc., Thousand Oaks, CA, USA. ⁸University of Nantes, Nantes, France.

Received: 12 October 2017 Accepted: 1 March 2018 Published online: 04 April 2018

References

 National Cancer Institute: SEER Cancer Statistics Factsheets: Myeloma. Surveillance, Epidemiology, and End Results Program website. http://seer. cancer.gov/statfacts/html/mulmy.html. Accessed 6 Oct 2016.

- Durie B, Moreau P, Sonneveld P, Morgan GJ, Lahuerta JJ, Beksac M, et al. Regional differences in the treatment approaches for relapsed multiple myeloma: an IMF study. J Clin Oncol. 2012;30(Suppl):Abstract 8095.
- 4. Jordan K, Proskorovsky I, Lewis P, Ishak J, Payne K, Lordan N, et al. Effect of general symptom level, specific adverse events, treatment patterns, and patient characteristics on health-related quality of life in patients with multiple myeloma: results of a European, multicenter cohort study. Support Care Cancer. 2014;22:417–26.
- Kumar SK, Therneau TM, Gertz MA, Lacy MQ, Dispenzieri A, Rajkumar SV, et al. Clinical course of patients with relapsed multiple myeloma. Mayo Clin Proc. 2004;79:867–74.
- Mols F, Oerlemans S, Vos AH, Koster A, Verelst S, Sonneveld P, et al. Healthrelated quality of life and disease-specific complaints among multiple myeloma patients up to 10 yr after diagnosis: results from a populationbased study using the PROFILES registry. Eur J Hematol. 2012;89:311–9.
- Orlowski RZ. Novel agents for multiple myeloma to overcome resistance in phase III clinical trials. Semin Oncol. 2013;40:634–51.
- Armoiry X, Fagnani F, Benboubker L, Facon T, Fermand JP, Hulin C, et al. Management of relapsed or refractory multiple myeloma in French hospitals and estimation of associated direct costs: a multi-centre retrospective cohort study. J Clin Pharm Ther. 2011;36:19–26.
- Garrison LP Jr, Wang ST, Huang H, Ba-Mancini A, Shi H, Chen K, et al. The costeffectiveness of initial treatment of multiple myeloma in the U.S. with bortezomib plus melphalan and prednisone versus thalidomide plus melphalan and prednisone or lenalidomide plus melphalan and prednisone with continuous lenalidomide maintenance treatment. Oncologist. 2013;18:27–36.
- Ghatnekar O, Alvegård T, Conradi N, Lenhoff S, Mellqvist UH, Persson U, et al. Direct hospital resource utilization and costs of treating patients with multiple myeloma in southwest Sweden: a 5-year retrospective analysis. Clin Ther. 2008;30:1704–13.
- Dimopoulos MA, Goldschmidt H, Niesvizky R, Joshua D, Chng W-J, Oriol A, et al. Carfilzomib or bortezomib in relapsed or refractory multiple myeloma (ENDEAVOR): an interim overall survival analysis of an open-label, randomised, phase 3 trial. Lancet Oncol. 2017;18:1327–37.
- 12. National Comprehensive Cancer Network website. https://www.nccn.org/. Accessed 15 Jan 2017.
- 13. Usmani SZ, Rodriguez-Otero P, Bhutani M, Mateos MV, Miguel JS. Defining and treating high-risk multiple myeloma. Leukemia. 2015;29:2119–25.
- Attal M, Lauwers-Cances V, Marit G, Caillot D, Moreau P, Facon T, et al. Lenalidomide maintenance after stem-cell transplantation for multiple myeloma. New Engl J Med. 2012;366:1782–91.
- Benboubker L, Dimopoulos MA, Dispenzieri A, Catalano J, Belch AR, Cavo M, et al. Lenalidomide and dexamethasone in transplant-ineligible patients with myeloma. New Engl J Med. 2014;371:906–17.
- Cerrato C, Gay F, Petrucci MT, Pellegrino M, Gaidano G, Offidani M, et al. Continuous treatment improves survival of newly diagnosed multiple myeloma patients achieving complete response: data from 5 phase III trials including young and elderly patients. Clin Lymphoma Myeloma Leuk. 2015; 15(Suppl 3):e54.
- McCarthy PL, Owzar K, Hofmeister CC, Hurd DD, Hassoun H, Richardson PG, et al. Lenalidomide after stem-cell transplantation for multiple myeloma. New Engl J Med. 2012;366:1770–81.
- Palumbo A, Bringhen S, Larocca A, Rossi D, Di Raimondo F, Magarotto V, et al. Bortezomib-melphalan-prednisone-thalidomide followed by maintenance with bortezomib-thalidomide compared with bortezomib-melphalanprednisone for initial treatment of multiple myeloma: updated follow-up and improved survival. J Clin Oncol. 2014;32:634–40.
- Palumbo A, Gay F, Cavallo F, Di Raimondo F, Larocca A, Hardan I, et al. Continuous therapy versus fixed duration of therapy in patients with newly diagnosed multiple myeloma. J Clin Oncol. 2015;30:3459–66.
- Sonneveld P, Schmidt-Wolf IG, van der Holt B, El Jarari L, Bertsch U, Salwender H, et al. Bortezomib induction and maintenance treatment in patients with newly diagnosed multiple myeloma: results of the randomized phase III HOVON-65/GMMG-HD4 trial. J Clin Oncol. 2012;30:2946–55.
- Musto P, Montefusco V. Are maintenance and continuous therapies indicated for every patient with multiple myeloma? Expert Rev Hematol. 2016;9:743–51.

- Dimopoulos MA, Moreau P, Palumbo A, Joshua D, Pour L, Hájek R, et al. Carfilzomib and dexamethasone versus bortezomib and dexamethasone for patients with relapsed or refractory multiple myeloma (ENDEAVOR): a randomized, phase 3, open-label, multicentre study. Lancet Oncol. 2015;17:27–38.
- Richardson PG. Update on proteasome inhibitors in multiple myeloma. Clin Adv Hematol Oncol. 2014;12:179–81.
- Siegel DS, Martin T, Wang M, Vij R, Jakubowiak AJ, Lonial S, et al. A phase 2 study of single-agent carfilzomib (PX-171-003-A1) in patients with relapsed and refractory multiple myeloma. Blood. 2012;120:2817–25.
- Stewart AK, Rajkumar SV, Dimopoulos MA, Masszi T, Špička I, Oriol A, et al. Carfilzomib, lenalidomide, and dexamethasone for relapsed multiple myeloma. New Engl J Med. 2015;372:142–52.
- Sugumar D, Keller J, Vij R. Targeted treatments for multiple myeloma: specific role of carfilzomib. Pharmacogenomics Pers Med. 2015;8:23–33.
- Wang M, Martin T, Bensinger W, Alsina M, Siegel DS, Kavalerchik E, et al. Phase 2 dose-expansion study (PX-171-006) of carfilzomib, lenalidomide, and low-dose dexamethasone in relapsed or progressive multiple myeloma. Blood. 2013;122:2284–90.
- Niesvizky R, Martin TG 3rd, Bensinger WI, Alsina M, Siegel DS, Kunkel LA, et al. Phase 1b dose-escalation study (PX-171-006) of carfilzomib, lenalidomide, and low-dose dexamethasone in relapsed or progressive multiple myeloma. Clin Cancer Res. 2013;19:2248–56.
- Siegel DS, Dimopoulos MA, Ludwig H, Facon T, Goldschmidt H, Jakubowiak A, et al. Improvement in overall survival with carfilzomib, lenalidomide, and dexamethasone in patients with relapsed or refractory multiple myeloma. J Clin Oncol. 2018; https://doi.org/10.1200/JCO.2017.76.5032.
- Collet D. Modelling survival data in medical research. 2nd ed. New York: Chapman & Hall; 2003.
- Simon R, Makuch RW. A non-parametric graphical representation of the relationship between survival and the occurrence of an event: application to responder versus non-responder bias. Stat Med. 1984;3:35–44.
- Durie BG, Harousseau JL, Miguel JS, Bladé J, Barlogie B, Anderson K, et al. International uniform response criteria for multiple myeloma. Leukemia. 2006;20:1467–73.
- Dimopoulos MA, Oriol A, Nahi H, San-Miguel J, Bahlis NJ, Usmani SZ, et al. Daratumumab, lenalidomide, and dexamethasone for multiple myeloma. New Engl J Med. 2016;375:1319–31.
- Lonial S, Dimopoulos M, Palumbo A, White D, Grosicki S, Spicka I, et al. Elotuzumab therapy for relapsed or refractory multiple myeloma. New Engl J Med. 2015;373:621–31.
- Moreau P, Masszi T, Grzasko N, Bahlis NJ, Hansson M, Pour L, et al. Oral ixazomib, lenalidomide, and dexamethasone for multiple myeloma. New Engl J Med. 2016;374:1621–34.
- Dimopoulos MA, Stewart AK, Rajkumar V, Masszi T, Oriol A, Hajek R, et al. Carfilzomib-lenalidomide-dexamethasone vs lenalidomidedexamethasone in relapsed multiple myeloma by previous treatment. Blood Cancer J. 2017;7(4):e554.
- Chanan-Khan A, Giral S. Complete response in multiple myeloma and the impact of novel agent. J Clin Oncol. 2010;28:2612.
- Van de Velde H, Liu X, Chen G, Xiangyang L, Gang C, Cakan A, et al. Complete response correlates with long-term survival and progression-free survival in high-dose therapy in multiple myeloma. Haematologica. 2007;92:1399–406.
- Lonial S, Anderson KC. Association of response endpoints with survival outcomes in multiple myeloma. Leukemia. 2014;28:258–68.
- Avet-Loiseau H, Attal M, Moreau P, Charbonnel C, Garban F, Hulin C, et al. Genetic abnormalities and survival in multiple myeloma: the experience of the Intergroupe Francophone du Myélome. Blood. 2007;109:3489–95.
- 41. Avet-Loiseau H, Soulier J, Fermand JP, Yakoub-Agha I, Attal M, Hulin C, et al. Impact of high-risk cytogenetics and prior therapy on outcomes in patients with advanced relapsed or refractory multiple myeloma treated with lenalidomide plus dexaméthasone. Leukemia. 2010;24:623–8.
- 42. Dimopoulos MA, Weisel KC, Song KW, Delforge M, Karlin L, Goldschmidt H, et al. Cytogenetics and long-term survival of patients with refractory or relapsed and refractory multiple myeloma treated with pomalidomide and low-dose dexamethasone. Haematologica. 2015;100:1327–33.
- Kurtin SE. Relapsed or relapsed/refractory multiple myeloma. J Adv Pract Oncol. 2013;4:5–14.
- 44. Jagannath S, Roy A, Kish J, Lunacsek O, Globe D, Eaddy M, et al. Real-world treatment patterns and associated progression-free survival in relapsed/ refractory multiple myeloma among US community oncology practices. Expert Rev Hematol. 2016;9:707–17.

- Song X, Cong Z, Wilson K. Real-world treatment patterns, comorbidities, and disease-related complications in patients with multiple myeloma in the United States. Curr Med Res Opin. 2016;32:95–103.
- Avet-Loiseau H, Fonseca R, Siegel D, Dimopoulos MA, Špička I, Masszi T, et al. Carfilzomib significantly improves the progression-free survival of highrisk patients in multiple myeloma. Blood. 2016;128:1174–80.
- Mateos MV, Richardson PG, Dimopoulos MA, Palumbo A, Anderson KC, Shi H, et al. Effect of cumulative bortezomib dose on survival in multiple myeloma patients receiving bortezomib-melphalan-prednisone in the phase III VISTA study. Am J Hematol. 2015;90:314–9.
- Jakubowiak AJ, Dytfeld D, Griffith KA, Lebovic D, Vesole DH, Jagannath S, et al. A phase 1/2 study of carfilzomib in combination with lenalidomide and low-dose dexamethasone as a front-line treatment for multiple myeloma. Blood. 2012;120:1801–9.
- Jakubowiak AJ, Dytfeld D, Griffith KA, Jasielec J, McDonnell K, Lebovic D, et al. Treatment outcome with the combination of carfilzomib, lenalidomide, and low-dose dexamethasone (CRd) for newly diagnosed multiple myeloma (NDMM) after extended follow-up. J Clin Oncol. 2013;31(Suppl): Abstract 8543.
- Korde N, Zingone A, Kwok ML, Manasanch EE, Bhutani M, Tageja N, et al. Phase II clinical and correlative study carfilzomib, lenalidomide, and dexamethasone followed by lenalidomide extended dosing (CRD-R) induces high rates of MRD negativity in newly diagnosed multiple myeloma (MM) patients. Blood. 2013;122:538.
- European Medicines Agency. http://www.ema.europa.eu/docs/en_GB/ document_library/EPAR_-_Summary_for_the_public/human/003790/ WC500197695.pdf. Accessed 24 Feb 2017.

Submit your next manuscript to BioMed Central and we will help you at every step:

- We accept pre-submission inquiries
- Our selector tool helps you to find the most relevant journal
- We provide round the clock customer support
- Convenient online submission
- Thorough peer review
- Inclusion in PubMed and all major indexing services
- Maximum visibility for your research

Submit your manuscript at www.biomedcentral.com/submit

